## Lawrence Berkeley National Laboratory

**LBL Publications** 

## Title

A framework to measure the technical, economic, and rate impacts of distributed solar, electric vehicles, and storage

Permalink https://escholarship.org/uc/item/896492dd

## Authors

Carvallo, Juan Pablo Bieler, Stephanie Collins, Myles <u>et al.</u>

## **Publication Date**

2021-09-01

## DOI

10.1016/j.apenergy.2021.117160

Peer reviewed



Electricity Markets & Policy Energy Analysis & Environmental Impacts Division Lawrence Berkeley National Laboratory

## A framework to measure the technical, economic, and rate impacts of distributed solar, electric vehicles, and storage

Juan Pablo Carvallo<sup>1</sup>, Stephanie Bieler<sup>3</sup>, Myles Collins<sup>3</sup>, Joscha Mueller<sup>2</sup>, Christoph Gehbauer<sup>2</sup>, Douglas J. Gotham<sup>4</sup>, and Peter H. Larsen<sup>1</sup>

<sup>1</sup> Electricity Markets & Policy, Lawrence Berkeley National Laboratory

- <sup>2</sup> Grid Integration Group, Lawrence Berkeley National Laboratory
- <sup>3</sup> Nexant, Inc.

<sup>4</sup> State Utility Forecasting Group, Discovery Park, Purdue University

## June 2021

This is a pre-print version of a journal article published in *Applied Energy*. DOI: https://doi.org/10.1016/j.apenergy.2021.117160



The work described in this study was funded by the Indiana Utility Regulatory Commission under Strategic Partnership Projects Agreement No. FP00009789.

#### DISCLAIMER

This document was prepared as an account of work sponsored by the United States Government. While this document is believed to contain correct information, neither the United States Government nor any agency thereof, nor The Regents of the University of California, nor any of their employees, makes any warranty, express or implied, or assumes any legal responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by its trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or The Regents of the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof.

Ernest Orlando Lawrence Berkeley National Laboratory is an equal opportunity employer.

#### **COPYRIGHT NOTICE**

This manuscript has been authored by an author at Lawrence Berkeley National Laboratory under Contract No. DE-AC02-05CH11231 with the U.S. Department of Energy. The U.S. Government retains, and the publisher, by accepting the article for publication, acknowledges, that the U.S. Government retains a non-exclusive, paid-up, irrevocable, worldwide license to publish or reproduce the published form of this manuscript, or allow others to do so, for U.S. Government purposes.

# A framework to measure the technical, economic, and rate impacts of distributed solar, electric vehicles, and storage

Juan Pablo Carvallo<sup>1,\*</sup>, Stephanie Bieler<sup>3</sup>, Myles Collins<sup>3</sup>, Joscha Mueller<sup>2</sup>, Christoph Gehbauer<sup>2</sup>, Douglas J. Gotham<sup>4</sup>, and Peter H. Larsen<sup>1</sup>

 <sup>1</sup> Electricity Markets and Policy Group, Lawrence Berkeley National Laboratory, 1 Cyclotron Road, MS 90R4000, Berkeley, CA 94720, United States
<sup>2</sup> Grid Integration Group, Lawrence Berkeley National Laboratory, 1 Cyclotron Road, MS 90R4000, Berkeley, CA 94720, United States
<sup>3</sup> Nexant, Inc., 49 Stevenson St Suite 700, San Francisco, CA 94105
<sup>4</sup> State Utility Forecasting Group, Discovery Park, Purdue University, West Lafayette, IN 47907

\*Corresponding Author (jpcarvallo@lbl.gov)

## Abstract

This study explores the joint impacts across the power system of distributed energy resources (DER) that could be deployed in utility distribution systems through an analysis of generation, transmission, and distribution expansion and costs driven by DER adoption. We identify six adoption scenarios that combine deployment levels of rooftop solar photovoltaic modules (PV), electric vehicle charging (EV), and battery storage in residential and commercial customers connected to representative feeders in Indiana by 2025 and 2040. Indiana is a good proxy for many U.S. states with low current DER adoption but potentially high future growth. The economic value of DER is assessed by developing capacity expansion and power flow analysis of the generation and distribution segments, respectively, under future hourly demand assumptions based on each adoption scenarios. Results for the distribution system power flow simulations show that voltage violations are relatively rare. Voltage violations can be mitigated at a very low cost using a combination of smart inverters in future rooftop PV systems and voltage adjustments in the feeder heads. Line loading issues are minimal, with only 0.2% of simulation hours showing loading levels above 100% of capacity. Generation capacity impacts are driven by unmanaged EV charging and could be mitigated with charging management. We estimate that the incremental rate impact from power system investment and operation of increased DER adoption in Indiana will be between -1.6% to +2% in 2025 and +0.2% to +15% in 2040 relative to the base case.

Keywords: distributed energy resources; power system; cost benefit analysis; rooftop solar

## List of abbreviations

ANSI	American National Standards Institute
BPS	Bulk Power System
СССТ	Combined Cycle Combustion Turbine
DER	Distributed Energy Resources
DG	Distributed Generation
DR	Demand Response
EE	Energy Efficiency
EV	Electric Vehicle
IOU	Investor Owned Utility
IRP	Integrated Resource Planning
LMP	Locational Marginal Prices
LTC	Load Tap Changer
0&M	Operations and Maintenance
PCA	Principal Component Analysis
PV	Solar photovoltaic
SCCT	Simple Cycle Combustion Turbine
T&D	Transmission and Distribution

## 1. Introduction

There are several types of emerging technologies that are being deployed or could be deployed in the distribution system and behind the meter. Technologies can produce electricity (e.g. solar photovoltaic (PV) panels, natural gas micro-turbines), store electricity (e.g. batteries, flywheels), consume electricity in novel ways (e.g. electric vehicles) and improve electricity management and consumption (e.g. smart thermostats, super-efficient appliances). These technologies are grouped and identified throughout this document as Distributed Energy Resources (DER). DER hold the promise to add value to the power system, but their decentralized deployment and operation will technically and economically impact different segments in the power system [1]. The motivation for this paper is to understand the impacts that DER adoption and operation will have across a regional power system. This paper answers the question: what are the technical and economic impacts of DER adoption in distribution, transmission, and generation? The paper answers the question by developing a framework to create DER adoption and operation scenarios whose output can be integrated into existing distribution and bulk power system (BPS) techno-economic simulation and optimization tools.

Over the last decade, customer-owned DER uptake has increased across the U.S. This increase has been driven by policies, prices, consumer attitudes, and attractive financing options for customers. Research oriented to understand the impacts of DER penetration has focused on U.S. states with higher adoption levels such as Hawaii, Massachusetts, or California. This paper uses the state of Indiana as a case study to produce results that would apply more generally to the vast majority of U.S. states that have yet to experience an increase of DER uptake. Based on the information shared by the five Indiana investor owned utilities (IOUs) that contributed to this study, only 0.14% of residential customers and 4.7% of commercial customers own a PV system, and almost no customers in Indiana own a storage system. Integrated resource plans (IRP) filed by Indiana IOUs shows an increased focus on customer-owned PV and EV adoption, but no analysis of DER battery storage (see Table A.1 in Supplementary Information(SI)). It follows that states like Indiana have significant room to expand DER adoption and would benefit from a planning framework and research findings that would help understand the impacts of these expansions. This work was developed at the request of the Indiana Utility Regulatory Commission, which explains the applied approach of integrating existing models to make this framework easier to understand and adopt by utilities, system planners, and other stakeholders.

This paper makes several novel contributions. First, it develops and applies a framework to assess impacts of DER adoption across the distribution, transmission, and generation systems. There are very few attempts to develop impacts of DER across all segments of the system, but they usually employ complex models that require substantial simplification of the grid, especially the distribution system. The proposed approach produces a coordinated sequence of simulations that produce technical and cost estimates for DER impacts using industry-standard models. Second, the paper produces empirically-derived adoption scenarios and allocates DER to customers based on income and consumption levels for realistic spatial distribution. The six scenarios developed are generalizable to any jurisdiction that wants to understand the interdependency of solar PV, electric vehicle, and energy storage adoption based on adoption and operational patterns. Third, it develops a rigorous technical impact analysis in distribution and generation using power flow and capacity expansion tools, then

estimates cost impacts, and finally translates these outcomes into rate impacts. There are no known papers that have addressed these three impact components in the same framework. The relevance of the joint analysis is two-fold. On one hand, it suggests that certain segments of the power system – distribution, transmission, and generation – can be benefited or unaffected by DER operation, while others can be substantially affected in the same scenario. An individual segment analysis is unable to capture this dependencies. On the other hand, it shows how technical and economic impacts are ultimately reflected in rates, whose levels depend on the balance of retail sales, peak demand, capital investments, and operational costs. A non-integrated analysis may find cost reductions that may, under certain circumstances, still lead to increases in rates. Finally, the paper develops a rigorous statistical method to cluster actual distribution feeders to produce a representative topology of feeders focused on capacity expansion analysis. Distribution systems are complex, but relatively homogenous within service territories, and hence it is possible to represent the aggregate impacts of millions of distribution system customers with a reduced number of representative circuits. While work on this space exists (e.g. [2–4]), this is the first application of representative feeders in a whole-systems analysis.

In the rest of the report, section 2 summarizes applicable literature, highlighting how this paper builds from and expands existing work. Section 3 details the methodology developed to represent the distribution system, develop scenarios, assess technical impacts, and estimate cost and rate impacts. Section 4 explains the approach to create DER adoption and operation scenarios for this application... Section 5 presents the technical impact results and section 6 presents the economic and rate impacts, both separately for each segment of the power system. Section 7 concludes with result and innovation highlights. All monetary values in this report are expressed in real 2017 dollars unless otherwise indicated.

## 2. Literature review

Several fields of study contribute to the growing body of literature examining the implications of increasing DER penetration in power systems. These studies explore current and future DER adoption trajectories and assess the impact across a number of dimensions, including the distribution system, bulk power system, distribution planning processes, ratepayer and societal costs and benefits, and utility business models. Table 1 organizes the scope and dimensions of analysis that characterize the work to understand DER impacts in the power system. In terms of the power system structure, the literature generally focuses on either the distribution or the transmission-generation (i.e. the BPS) segments, with much limited work on all three jointly given computational challenges. In terms of dimension of analysis, work can be categorized as focused on technical or physical results, economic results, and in fewer cases rate impacts. As shown in Table 1, each one of these combinations can include several papers. Additionally, papers tend to focus on single DER (e.g. solar PV, distributed generation, or EV) but rarely examine several at the same time. An exhaustive analysis of work in all combinations in Table 1 – essentially, on each cell in the matrix – is impractical. The main purpose of Table 1 is to demonstrate that there is little to no work that encompasses all these cells at the same time and for several different DERs. The framework to achieve this analysis in a coordinated, sequential manner utilizing industry-standard tools is the main contribution of this paper. In this review we focus on research on (1) technical and economic impacts of DER on distribution system, (2) technical and

economic impacts of DER in the BPS, and (3) joint analysis of distribution and BPS impacts. This selection is justified because research in these topics is used to inform the tools and approaches employed in this paper and to focus on the few whole-system analyses that exist and how the present work differs from them.

Table :	1. Scope	of literature	reviewed
---------	----------	---------------	----------

	Dimension of analysis						
Power system segment	Technical/Physical	Economic (cost and/or benefit)	Rates				
Distribution	[5], [6], [7], [8], [9], [10]	[11], [12], [13], [14] (Value of solar)	[17,18]				
Transmission and	[19], [20], [21]	[15,16] (BPS)					
Generation							

A number of studies have modeled high PV penetration on feeders and assessed the technical impacts. A paper found that distributed generation (DG) can have positive impacts (voltage support, deferred capital investments) and negative impacts (protection coordination, voltage regulation, voltage flicker, short circuit levels) [5]. The Pacific Northwestern National Laboratory (PNNL) summarizes the major types of analysis conducted on electric distribution systems along with their applications and relative maturity levels [6]. Special emphasis is placed on distribution system analyses required for increasing levels of DERs. The National Renewable Energy Laboratory (NREL) catalogs distribution-level impacts of high PV penetration, including overload-related, voltage-related, reverse power flow, and system protection impacts [7]. EPRI discusses practical planning limits for adding DG to distribution circuits [8]. The report classifies the limits into four categories: voltage regulation (e.g. voltage rise), rapid voltage change (fluctuations, sudden loss of generation), thermal limits (capacity, losses), and protection limits (overcurrent, islanding) Another paper investigates the impact of different electricity pricing systems and their impact on charging of electric vehicles (EVs) and battery storage [9]. It then compares the impact EVs and storage would have in the different scenarios on the integration of photovoltaic (PV) systems. However, neither of these studies analyze the technical impacts on transmission and generation.

A growing body of literature analyzes the benefits and costs of DER. [11] reviews methods for analyzing the benefits and costs of distributed PV generation to the U.S. electric utility system. Utilities will occasionally commission "value of solar" studies in their service territories to understand the benefits and costs specific to their geographic location, generation portfolio and customer base. RMI reviews sixteen distributed PV benefit/cost studies by utilities, national labs, and other organizations [12]. These studies reflect a significant range of estimated distributed PV value. Some studies examine costs and benefits at a broader level, such as the economic impact of distributed PV in California [13] and, closer to Indiana, the value of DG in Illinois [14]. Elements of these studies inform the DER valuation framework developed in this paper. It is important to note that these studies solely focus on distributed solar, not capturing the interactions with other DER such as the ones analyzed in this paper. In addition,

they tend to significantly simplify the simulation of impacts in the distribution system by using historical investment trends, as opposed to the feeder-level analysis proposed in this paper.

Several studies have addressed the impacts of DER on the BPS. The Electric Reliability Council of Texas (ERCOT) identified areas of concern related to reliability impacts of DER to the BPS: increased error in load forecasting, less accurate inputs to Independent System Operator (ISO) functions, and uncoordinated system restoration after a load shed event [19]. The North American Electric Reliability Corporation (NERC) examined the potential reliability risks and mitigation approaches for increased levels of DER on the BPS. The objective was to help regulators, policy makers, and other stakeholders better understand the differences between DER and conventional generation with regards to the effect on the BPS [20]. NERC also created a DER Task Force which developed DER modeling recommendations for BPS planning studies [21]. There are techno-economic analyses investigating the impact of DERs on transmission costs and investments using an approach to optimize the location of DERs to minimize costs [15,16]. While these techno-economic studies are useful, they do not reflect the impacts of unmanaged DER location that characterize existing distribution systems.

Whole system studies have been framed in different ways. A strand of literature concerns with integrating energy and electricity delivery systems. For example, heat and electricity systems were analyzed for the U.K. [22] and Germany [23], including generic national optimization frameworks [24]. A few studies have investigated the impacts of DER on the distribution and bulk power system in an integrated or combined analysis. One study utilized synthetic distribution grid models combined with a transmission grid model to represent the German transmission grid to generate an integrated model of the distribution and bulk power system [25]. Another study did create an integrated model to analyze reactive power management in an approach considering the transmission and distribution level. This was also based on the German transmission grid and combined with generic distribution grid models [26]. The present paper, however, focuses on a sequential analysis of all electricity system segments: distribution, transmission, and generation. In this space, papers have developed whole-system modeling efforts to jointly optimize investment in generation, transmission and distribution. For example, the techno-economic effect of bulk and distributed energy storage on the power system of Great Britain were analyzed [27]. Another example discussed investment decisions in DER vs distribution system expansion in the context of electricity access [28] and for a mature U.S. vertically integrated utility [1]. A recently developed model was built with enhanced capacity to handle vast amounts of data to simulate distributed and BPS resources [29]. These integrated models generally simplify distribution system simulation and use synthetic distribution system models for their analysis rather than real world feeder models, which ignore the unique configurations of some real world feeders. These large integrated models are useful for coordinated planning, but this only applies to certain vertically integrated utilities. In contrast to a single integrated model, this paper suggest methods to produce and interpret results of system-specific models – BPS and distribution – to assess the impacts of DER across the whole system, regardless of their economic organization [30].

## 3. Methodology

Figure 1 describes the structure of this study in terms of four key processes. First, the distribution system is characterized by identifying representative feeders and creating scaling factors for each.

Second, DER adoption and operation scenarios are developed for each customer on each representative feeder. Their net load is scaled and aggregated to the state level for the transmission-generation analysis. Third, a techno-economic capacity expansion and dispatch analysis is developed for the distribution system and the BPS using the industry-standard Cymdist and Aurora models, respectively [31]. This analysis is based on the two previous steps: representative feeders and customer net load for distribution system power flow and transmission-level demand for generation-transmission expansion and dispatch. Finally, retail sales, peak demand, and cost impacts are integrated into a ratemaking model to assess rate impacts.



Fig. 1. Process and stages included in the methodology

DER can impose technical costs to the distribution system due to their impact on voltage levels and line loading, among other impacts. DER can also benefit the distribution system by reducing line and transformer losses and by deferring capacity investments. Due to the integrated nature of power systems, DER costs and benefits can also accrue in the transmission and generation levels. We call these economic outcomes of DER integration "value streams". DER have a wide array of value streams [32–34], but this study focuses on a subset of possible value components including energy cost, losses, and capital deferment (capacity value). Due to technical and resource limitations, a number of additional value streams identified in the literature were not considered. These include DER impacts on ancillary services, fuel price hedging, and wholesale price reduction. Ancillary services such as frequency regulation can be a relevant value stream for battery storage [35]. However, there is no simplified method to determine the potential contribution of DER to this value stream that could be applied

within our framework.

## 3.1 Representative feeders

In contrast to most integrative studies of DER impacts, we sought to rigorously capture the impact of DER on distribution system through power flow simulations. The sheer number of feeders in Indiana require a sampling approach to identify representative feeders that can be used for simulation and result extrapolation. This study employs clustering techniques, especially the method developed by [2], to produce a set of six representative feeders for the five IOUs in the state of Indiana. The number of representative feeders balances breadth with the ability to accommodate the number of power flow simulations necessary given the number of customer loads, time horizons, and DER adoption scenarios. On average, each feeder is simulated approximately 570 times.

The method to identify representative feeders involves four steps: (1) identify available feeder metrics for clustering, (2) transform the data using Principal Component Analysis (PCA) and identify outliers, (3) determine the optimal number of clusters, and (4) select representative feeders for each cluster. After feeders are selected, DER adoption and synthetic hourly operation are simulated for each customer on each feeder. A detailed explanation of the feeder sampling, clustering, and selection process, and input data creation for power flow runs is available in section C in the SI. Since the feeder clustering process is considered part of the method – rather than a result of this study – a description and key statistical variables for the final representative feeders with are shown in Table 2.

Cluster	General description of feeders in cluster	Average customer number	Average total length (miles)	Average CAIDI (min)	Share of installed capacity (residential)	Share of installed capacity (commercial)	Share of installed capacity (industrial)	Share of circuit length that is underground
	Short and high commercial,							
1	about 1/3 underground	445	9.5	145.1	25%	58%	6%	30%
2	Short, urban residential Suburban mostly overhead,	567	11.5	142.4	77%	17%	2%	19%
3	residential, relatively dense Very long residential mostly	1,472	21.7	135.4	70%	21%	7%	20%
4	rural Suburban underground	1,133	59.3	148.5	78%	15%	3%	19%
5	residential relatively dense Short, heavy industrial,	1,535	26.2	121.4	77%	17%	5%	67%
6	substantial underground	463	10.0	120.8	15%	31%	51%	39%

Table 2. Six representative clusters and a sample set of parameter statistics

## 3.2 Energy costs

Operation of DER changes the shape and level of the net demand that is supplied by the BPS. The change in shape can produce costs or benefits depending on how the BPS dispatch curve changes and whether more flexible resources for ramping are needed (e.g. to address the "duck curve" phenomenon) that would incur additional fuel charges.

Change in levels can also be bidirectional: net demand can decrease with high levels of PV generation, often resulting in savings from less energy produced at the utility-scale. However, BPS energy

consumption can also increase with EV charging. The timing of these changes, captured by the shape component, impacts resource adequacy requirement at the BPS-level. However, these capacity requirements are captured through a different value stream described later.

Changes in energy consumption and their monetization will employ the Aurora capacity expansion and production cost model. The process to employ Aurora for this purpose follows these steps:

- 1. Produce hourly net demand differentials between the base case scenario and each one of the five adoption scenarios presented in Section 4.
- 2. Add the scenario net demand differentials to the base case to produce five net demand sets that are consistent with their assumptions, but at the same time reflect the adoption levels determined in this study's scenarios.
- 3. Interpolate the years between 2025 and 2040 to provide the data needed for the capacity expansion model.
- 4. Input these assumptions in the model and run it for each hour of the year.
- 5. Calculate the dispatch costs (fuel and non-fuel variable costs, ramping costs, and spinning and non-spinning reserves costs) for each hour, and produce annual totals.
- 6. Compare state-wide present value of dispatch costs for each adoption scenario against the base case.

## 3.3 Losses

Transmission and distribution losses may be reduced or increased due to the presence of DER. Distribution losses can go in either direction depending on their capacity relative to the hosting capacity and their location within the feeder. Traditionally, distribution feeders follow a "conic" construction method, with higher gauge wire close to the head and lower gauge wire close to the ends. Then, higher power flow levels close to the end of the feeder have a disproportionate impact on losses compared to the same flow levels close to the feeder head. Transmission losses would generally decrease due to reduced loading in the lines. For the purposes of this study, we do not assume that DER deployment results in power flowing back into the transmission system with a corresponding increase in losses.

Distribution line losses for the primary voltage system will be assessed directly from the Cymdist modeling results for each representative feeder. We will prepare and run a specific set of simulations for energy losses using 24 hours on a typical day per season (fall, winter, spring, and summer). The days are selected as the median load day on each season. The objective of this approach is to capture typical losses levels that are representative of the adoption scenario, rather than losses at maximum/minimum load conditions. Feeder-level energy losses levels for each scenario will be compared against the base case. Losses differences can then be monetized using either retail rate or an average wholesale purchase price.

Transmission losses cannot be directly calculated because there is no explicit modeling of the transmission system available. We will estimate transmission losses changes based on the difference between aggregate net demand in the base case scenario and the adoption scenarios. For example, if energy consumption is 10% higher in one adoption scenario compared to base case, then we will

assume that transmission system losses will be 10% higher as well. While imperfect, this will allow to monetize transmission losses changes into rates.

## 3.4 Capital deferment (capacity value)

DER operation can defer or increase future investments in generation, transmission, and distribution. As with losses reductions, DER may produce capital deferments in generation and transmission. However, DER deployment can require flow capacity and safety upgrades in the distribution system and can trigger the need for flexible resources at the generation and transmission level to meet additional ramping requirements.

## Generation

Capacity value of DER for the generation system can be directly calculated using the results from the Aurora capacity expansion model. Typically, most studies estimate the capacity credit of the different DER technologies, accounting for T&D losses (i.e. referring the capacity credit to the transmission network). However, since the Aurora model is able to simulate capacity expansion for different net demand scenarios, we can directly compare the adoption scenarios against the Base scenario to determine the difference in resource type, capacity mix, and cost.

We estimate potential reductions in planning reserve margin that come from peak demand reductions as part of the generation capacity value. We will implement a simple method that values the changes to the reserve margin based on the reserve requirement output from the Aurora model.

## Transmission

Transmission expansion costs are complex to estimate because of the bulky nature of transmission investments and the spatial distribution of transmission system lines and substations. The NREL study proposes three methods to assess capital deferments in transmission systems. Two of these methods require explicit modeling of the transmission network, which is out of the scope of the Comprehensive Study. The third method proposes obtaining transmission locational marginal prices (LMP) and determining the marginal contribution of DER to reduce those LMP. This reduction serves as a proxy for transmission capacity values. However, this method assumes that DER penetration levels do not substantially change the underlying LMP data used for the estimates. This assumption can produce large distortions when applied on analysis performed over long time frames such as this study's.

We developed a simplified method that involves linearizing transmission expansion by estimating a cost of transmission per peak MW transported. These costs are estimated using the rate base information separated by functional category.

#### Distribution

The methods to assess impacts of DER on distribution system vary significantly in complexity and outcomes. Given that this is a focus of the study, we implement a more sophisticated method based on power flow simulation of actual primary voltage feeder and load data as indicated earlier in this Section. This method has three parts. First, we run power flow simulations for each representative feeder for several combinations of adoption scenarios, hours of the year, and horizon (2025 and 2040). Second, we analyze the technical outcome of each power flow simulation by tracking voltage levels per

node, line losses, and line loading. These three parameters are drivers of the feeder upgrades. Finally, we scale feeder upgrades for each cluster to the whole cluster level, and then estimate state-wide DER distribution system integration costs and benefits.

Simulations are performed on the Cymdist power engineering software from CYME/Eaton. Cymdist has a Python API that is used to automate simulations<sup>1</sup>. All active and reactive loads from each Cymdist feeder model are overwritten by reading a csv file with pre-determined hourly values based on the Cymdist input data explained in section C.3 in the SI. The automated framework allows executing thousands of simulations within a short period of time.

We assume that feeders will be upgraded, if needed, to maintain voltage drop, line and transformer loading and losses, within prescribed and accepted levels. In some cases, the DER scenarios may be such that they will prevent an upgrade that would otherwise be required in the base case, accruing savings to the system. This means that we will estimate upgrades required for the base case and determine a total cost for a representative feeder. We then compare these reference costs against the costs to maintain the representative feeders for other adoption scenarios. The cost differential is the DER integration value, which could be positive (a cost) or negative (a savings).

There are no trustworthy automatic upgrade algorithms for distribution systems that can be applied to our setting [36]. Given the volume of simulations performed (close to 1800 individual power flows), we select certain scenarios, years, and hours of the year to manually inspect each representative feeder and decide to implement the following strategies to correct technical issues with feeders:

- Repowering conductors (line loading and losses)
- Add a new voltage regulator or modify the setting of an existing voltage regulator (voltage regulation)
- Modify a substation's tap changers (voltage regulation)
- Adopt and calibrate smart inverters for DER PV (voltage regulation)

Finally, distribution-level capital investments or deferments will be monetized based on current infrastructure costs that were provided by the three Indiana utilities whose feeders were used as the basis of this analysis (see section B in SI).

## 3.5 Rate impacts

The methods developed in subsections 3.2, 3.3 and 3.4 produce cost estimates for energy, losses, and capacity in generation, transmission, and distribution systems due to DER adoption. We calculate aggregate energy consumption by utility and year and pass this information along with the DER value changes to the ratemaking model. As stated in [37], "the [ratemaking] models determine annual revenue requirements based on each utility's costs associated with existing and future capital investments, operational expenses, debt, and taxes. Those costs are then allocated to the customer sectors and rates are determined using the annual energy forecasts." This is the same ratemaking

<sup>&</sup>lt;sup>1</sup> The Cymdist power flow simulations were performed using models and Functional Mockup Units developed during the DOE-funded project "CyDER: A Cyber Physical Co-Simulation Platform for Distributed Energy Resources in Smart Grids", which delivered a co-simulation platform based on the Functional Mockup Interface standard.

model employed by the Indiana Utility Regulatory Commission (IURC) in their proceedings.

## 4. Scenarios

This study produces six scenarios based on different levels of DER and EV adoption. The scenarios were developed to explore how the distribution system would perform under different DER adoption and demand levels. DER and demand are characterized across three dimensions: PV adoption, battery storage, and system demand. Each dimension has one of three adoption levels: business as usual (BAU), high, and very high. The scenarios cover two horizons: a short-term horizon (2025) and a long-term horizon (2040). See Table 3 for the quantitative details of the adoption levels in 2040.

Adoption Level	PV	Storage	Electric Vehicles	System Demand
BAU	Established from base case forecast from	Established from base case forecast	Established from base case	Established from base case forecast
	utility IRPs.	from utility IRPs.	forecast from utility IRPs.	from utility IRPs.
High	15% of customers by 2040 (Based on scenario from IPL IRP)	1% of customers by 2040	23% of vehicle stock by 2040 (Based on scenario from MISO Study)	Base Demand + EV addition
Very High	25% of customers by 2040 (Extrapolation of High Scenario)	5% of customers by 2040	68% of vehicle stock by 2040 (Based on scenario from MISO Study)	Base Demand + EV addition

Table 3. Quantitative scenario adoption levels in 2040

The purpose of these scenarios is to represent a set of possible futures to explore the behavior of the power system under different circumstances. This type of scenario exploration can help to identify situations in which the system may perform poorly and thus inform decision-makers. Given the applied nature of this work, scenario development follows a more intuitive approach that balances political sensitivity, technical feasibility, and data-driven quantitative approaches. The scenarios are policy-agnostic in that there is no assumption whether the DER adoption or system demand levels are attained through a particular policy mechanism.

The PV and storage dimensions for each scenario reflect the adoption of behind-the-meter DER by customers—and not utility-scale solar or storage. PV systems would thus be customer-installed rooftop PV for residential and commercial customers. Battery storage systems are less common than PV in each scenario and are assumed to be installed at the same site as PV. The batteries were sized to reflect the capacity of a system on the customer side of the meter and did not include any utility-scale batteries. The levels of system demand are driven by the adoption of electric vehicles. While a number of factors could arise to impact system demand, EVs are the most likely option for large-scale changes and provide a means to simplify scenario development. Table 4 shows the six scenarios:

Table 4. DER adoption scenarios

Scenario name	Description
Base	Represents the base case scenario. Each scenario dimension (PV, battery storage, and system demand) are taken from the base case scenarios of the utility IRPs. Note the distinction between "Base" to refer to this scenario and business-as-usual (BAU) to refer to the specific DER projection level as in BAU, high, and very high (see Table 2.2).
High electrification	Represents a scenario where system demand increases beyond base case projections, but DER adoption does not. This allows the analysis to explore the behavior of the distribution system in the case of high EV adoption—but with a configuration that reflects BAU levels of DER penetration.
High PV	Tests the scenario where PV adoption increases beyond BAU projections, but without large- scale additional system demand and without a large increase in battery storage adoption. Battery storage can mitigate some of the integration challenges for the utility of high rooftop PV penetration and this scenario tests the ability of the grid to handle more PV without the customer-side storage.
High PV and battery storage	Examines a scenario where a high level of rooftop PV penetration is coupled with a relatively high penetration of battery storage systems. The scenario assumes some breakthrough in battery technology, financing, and/or policy that would boost adoption, as current levels are close to zero. Even at a 'high' level, only 1 percent of customers adopt batteries. In this scenario, all battery storage systems are co-located with rooftop PV—though many rooftop PV systems are installed without batteries due to high PV penetration.
Battery storage arbitrage	Reflects a scenario where a storage breakthrough occurs, achieving a 'high,' 1 percent penetration level, with BAU levels of rooftop PV adoption. This scenario allows exploration of the impact of higher-than-expected battery storage adoption, while holding other factors at the baseline level.
Boundary case	Extrapolates adoption of rooftop PV, battery storage, and EVs to 'very high' penetration trajectory levels. The purpose of this scenario is to act as a boundary case and test the behavior of the distribution system with stressors that are beyond even the 'high' project levels. The 'very high' adoption levels are not present in any other scenarios.

Fig. 2 depicts the expected Indiana DER installed capacity for each scenario in 2040 (2025 is available in Fig. A.1 in the SI). These charts present nameplate installed capacity of each DER rather than coincident peak capacity. Residential and commercial PV adoption are relatively low in the Base and High Electrification scenarios, but up to 2 GW are deployed in the High PV scenarios. Battery storage adoption is relatively small across scenarios, ranging from 1 MW in the base scenario to almost 1 GW in the Boundary scenario. EV capacity is broken out by the type of charger used for the vehicle. Type 1 EV charges use less load at any given time, but take longer to charge, while Type 2 EVs charge quickly and use more load during a given hour. As a result, Type 1 EV makes up 21% of EV customers but only 11% of EV capacity. EV penetration ranges from 1.7 GW of charging capacity by 2040 in the Base scenario to over 12 GW of charging capacity in the Boundary scenario. The Boundary scenario then acts as a stress-test case to analyze the behavior of the distribution system with adoption levels beyond the most optimistic existing adoption scenarios. It is important to highlight that EV charging may occur at different times of day while PV injections across the Indiana territory will be highly correlated. This

means that the coincident hourly impact of PV may be higher than that of EV, even in scenarios where the latter has larger installed capacity.

These adoption patterns translate to varying net consumption outcomes at the state-wide level (see Figs. A.2 and A.3 in SI). By 2040, there are comparatively larger changes in annual consumption, with overall consumption levels increasing or decreasing depending on the scenario. The high electrification and boundary scenarios have relatively high levels of EV adoption and annual consumption for residential customers increase by 8% and 21%, respectively, compared to the base case. The scenarios with BAU EV adoption and high PV adoption show a 10% decrease in residential annual consumption and an 8% decrease in commercial annual consumption compared to the base case. The annual consumption for industrial customers does not vary by scenario because they are not considered DER adopters in this study. Industrial customers, however, make up a large portion of Indiana's overall consumption, accounting for 46% of total annual consumption in 2025 and 45% of total annual consumption in 2040.





Impacts during specific hours of the day exist even though on an annual basis net demand may not change substantially as a result of additional DERs. Solar PV and EV charging, for example, offset each other on an annual basis, but solar production and EV charging generally happen at different times during the day. Therefore, there are potentially large changes on an hourly level on each scenario. One of these changes is a shift in the hours that have the highest load concentration on peak days. Fig. 3

illustrates the timing of state-level aggregate peak day usage for each scenario. The plot shows the average hourly loads over the top ten peak load days in 2040 for each scenario. We can compare scenarios by displaying the percentage of usage in each hour (the area under each curve adds up to 100%). Peak days occur during summer months in all scenarios, but the peak hour changes depending on which DER is dominant. For scenarios with high levels of solar penetration, the peak hour tends to occur later in the evening, between 6-7 pm (hour 19 on the plot). For the Base, High Electrification, and High Storage scenarios the peak occurs earlier in the day, between 3-4 pm (hour 16 on the plot). The Boundary scenario, with large PV and solar penetration, shows a high concentration of load in the evening hours, with load from 6-9 pm, accounting for more than 30% of the daily load on peak days.

In addition to changes in peak load concentration, the magnitude of the annual peak as compared to the Base Case also changes. The High Electrification and Boundary scenarios have relatively high levels of EV adoption and peak demand increases by 17% and 83%, respectively. The scenarios with BAU EV adoption and high PV adoption show a 6% decrease in peak demand.





Finally, demand response (DR) and energy efficiency (EE) availability across scenarios was based on IRP forecasts provided by the IOUs. Unlike the other DERs, customer participation in EE and DR programs are largely driven by utility efforts. Therefore, EE and DR adoption were not varied in the above scenarios. For both 2025 and 2040, Indiana is expected to have a total DR capacity of almost 750 MW and total annual EE savings of almost 1,900 GWh.

## 5. Technical impact of DER on the distribution and generation systems

This section reports the technical impact analysis for the distribution and generation systems.

## 5.1 Distribution system technical impacts

This section reports the Cymdist power flow results on the distribution system for the three key output variables: voltage regulation, line loading, and line losses. Simulations are executed on each of the six representative primary voltage feeders, and characterized by (1) the year of analysis (2025 or 2040), which drives load growth and adoption levels; (2) six adoption scenarios, which establish different combinations for adoption levels of distributed PV, storage, and electric vehicles; and (3) twenty-four hours (a full day) on the minimum and maximum load days, for a total of 48 hours per feeder-scenario-year-cluster combination. These variables result in 576 power flow simulations per representative feeder, for a total of 3,456 simulations.

#### 5.1.1 Voltage regulation

Results for voltage regulation are reported in Fig.4 (next page). In this figure, the column panel reports the six adoption scenarios while the row panel depict the six representative feeders identified by their cluster CL1 to CL6. The charts show the statistical distribution of voltage in p.u. (per unit or the fraction of nominal voltage) for each simulated node-hour<sup>2</sup>, where the red shade represents 2025 and the blue shade 2040. The vertical lines represent the two ANSI voltage violation criteria: orange for the optimal range and red for the acceptable range. The y-axis is normalized to the highest value for each cluster to compare statistical distributions across scenarios for a given cluster, rather than across clusters.

At first glance, voltage violations are rare and minimal. About 0.5% of the node-hours simulated are under the 0.975 p.u. lower voltage range for the optimal scenario and 0.3% node-hours are above the 1.05 p.u. upper range. Only 0.04% of the node-hours are under the 0.95 p.u. lower acceptable range, and none are above the 1.058 upper acceptable range. The absolute minimum and maximum voltages are reasonably close to the ANSI limits for all of the node-hours simulated (Table 5). High voltage violations are very small, exceeding the optimal range by 0.009 p.u. in the worst case. Low voltage violations are also very limited, with a worst case excursion 0.053 p.u. below the optimal limit. Low load day simulation hours fall almost entirely within optimal and acceptable ranges; the majority of the voltage violations occur during high load days.

		Voltage Levels (p.u.)							
	Type of Load		25 <sup>th</sup>		75 <sup>th</sup>				
Year	Day	Minimum	Percentile	Median	Percentile	Maximum			
2025	High	0.957	1.009	1.02	1.033	1.054			
2025	Low	0.988	1.016	1.025	1.038	1.046			
2040	High	0.908	1.008	1.019	1.031	1.058			
2040	Low	0.945	1.016	1.026	1.038	1.057			

Table 5. Ranges in voltage regulation for low and high load day simulated hours, by year

<sup>&</sup>lt;sup>2</sup> A node-hour is a unique observation for a node on one of the 48 simulated hours. We treat node-hours as a single variable to be able to show results for the same node across different hours in the same chart, and avoid one additional dimension in the visualization.



Fig. 4. Distribution of voltage regulation by node-hour

However, it is important to study power systems in the extreme, because critical issues can be lost in a simple analysis of averages. We find optimal range voltage violations in at least one feeder node on 159 of the 3,456 simulated hours and acceptable range violations in 17 simulated hours (see Table 6 for optimal range violations). Representative feeders in clusters 1, 5, and 6 exhibit voltage violations, but only in the Boundary scenario. Cluster 2 feeder has two to three simulation hours with violations in the Base, High PV, High PV and Storage, and Storage scenarios; and five hours in the Boundary and High Electrification scenarios. Feeders for clusters 3 and 4 – among the longest in the sample – have the highest number of hours of voltage violations. In cluster 3, almost 20% of the simulated hours in the Boundary Case show voltage issues.

Table 6. Number of simulation hours with ANSI optimal range voltage violations by cluster and scenario for 2025 and 2040

	Scenario									
Cluster	Base	High Electrification	High PV	High PV and Battery Storage	Storage	Boundary Case				
CL1	0	0	0	0	0	9				
CL2	3	5	2	2	3	5				
CL3	9	8	11	11	9	19				
CL4	11	11	6	6	11	9				
CL5	0	0	0	0	0	8				
CL6	0	0	0	0	0	1				

This overview of voltage violation results suggests that some representative feeders are much more impacted by DER adoption than others and that the impact produces both low and high voltage issues. A detailed cluster by cluster analysis is included in section D in SI.

#### 5.1.2 Line loading

Cymdist calculates the percent loading of each line segment for each simulated hour, based on the line segments' capacity and power flow solution for a specific hour. Results for line loadings are reported in Fig. 5. In general, lines loading issues are non-existent in the short-term (2025) and minimal in the long-term (2040). Loading issues in 2040 arise in the Boundary scenario for clusters 3, 4, and 5 and in the High Electrification scenario for cluster 4.

Only eight simulation hours out of 3,456 simulated hours have overloaded line segments. In these eight hours, between 0.4% and 8% of line segments are overloaded, depending on the cluster (see Table 7). Line overloading takes place in very specific times of day, coinciding with peak residential demand (2-3 pm) or with DER PV production decline coupled with EV charging (6-7 pm). Overloading is also incremental, which means that mitigating the overload for the worst case scenario in each cluster (6 pm at each cluster) will also mitigate issues for the other simulated hours in the same cluster.



Fig. 5. Distribution of line loading by node-hour

Table 7. Simulation mours with overloading issues	Table 7.	Simulation	hours with	overloading	issues
---	----------	------------	------------	-------------	--------

Scenario	Cluster	Hour of Day	Number of Overloaded Segments	Number of Total Segments	Share of Overloaded Segments (% total)
Boundary	CL3	6 pm	31	592	5.2%
Boundary	CL3	7 pm	28	592	4.7%
Boundary	CL4	6 pm	15	1,621	0.9%
Boundary	CL4	7 pm	12	1,621	0.7%
High Electrification	CL4	2 pm	10	1,621	0.6%
High Electrification	CL4	3 pm	10	1,621	0.6%
Boundary	CL5	6 pm	43	535	8.0%
Boundary	CL5	7 pm	42	535	7.9%

Adoption of distributed PV has a beneficial effect in line loading. In the scenarios with higher PV adoption (High PV and High PV with Storage) the worst case line loading is typically 10%-15% less than the High Electrification scenario (with high EV adoption) and 5% less than the BAU scenario (with very little PV adoption). In contrast, it is likely that electric vehicle charging is leading to overloading issues across clusters because the timing of some of the overloading issues coincide with residential type I charging operations.

In summary, these results suggest that the existing capacity of line segments in representative feeders would be enough to accommodate the DER deployed under even the most stringent adoption scenario. Required re-conductoring expenditures should be relatively small considering the DER adoption levels. These costs are discussed in Section 6. Note that this analysis does not cover distribution transformer loading or secondary network loading. It is possible that these two components do not have the flexibility that the primary distribution system has and would therefore accrue additional DER integration costs.

It is important to highlight limitations of the loading analysis performed in this study. Annual energy consumption by customer was allocated using aggregate load profiles for the three customer types provided by some IOUs due to the lack of actual hourly load shapes or peak power consumption for each customer. The simulation results show that even in the Base case some line segments experience very high loading, while the average line segment loadings are around 5 percent for most clusters and scenarios. These high line loadings may have originated in the method of load allocation, which does not reflect actual customer peak loads. On one hand, the method employed may result in a higher peak load for customers with a relatively flat load demand. On the other hand, the method may produce lower peak load (and lower line loading) for customers with a volatile load profile.

#### 5.1.3 Line energy losses

Unlike voltage violations and overloading issues, line energy losses do not translate to power quality issues for customers. However, utilities monitor line losses to maintain a level that is cost-effective for the utility as well as their customers. This means that there is no set standard or benchmark for assessing an acceptable limit for line losses as this cost-effectiveness test will vary across utilities and

over time. Consequently, we focus on measuring change in losses between the Base scenario and the other scenarios as a measure of the differential impact of DER adoption.

Feeder losses for the highest hour of the year may be several times higher than average losses calculated using annual aggregates, but it is the latter that informs the overall economic impact of increases or reductions in energy losses. For this reason, we develop a special set of simulations depicting typical conditions in four seasons of the year that are more conducive to aggregate estimates. We select one day per season to capture seasonal patterns in demand and solar PV production. We report hourly losses to show the variation of losses throughout the day and how they correlate with specific DER usage patterns.

Feeder losses for 2025 and 2040 are reported in Fig. A.4 and Fig. A.5 in the SI, respectively. We report the average percentage change in feeder losses for three selected adoption scenarios: High Electrification, High PV, and Boundary relative to the Base case. We use these three scenarios because results for the High PV and Storage and Storage scenarios are almost identical to the results for the other scenarios and the Base case, respectively.

Results show that line losses follow the new patterns of net demand that arise with PV and EV adoption. Losses are higher than the Base case during the times of day when EV is charging (High Electrification scenario, between 2 pm and 7 pm). Losses are lower than the Base case in the scenarios with higher PV penetration and during the hours of PV production between 10 am and 4 pm. The Boundary scenario in years 2025 and 2040 shows the highest variation in losses compared to the Base case, because it includes very high PV and EV adoption levels. Losses can be 10 to 13 times higher than the Base case during peak demand hours with substantial amounts of residential EV charging.

We estimate annual average losses by calculating the average for each cluster and scenario across all simulated hours for all seasons (see Table A.2 in SI). Average results show moderate increases or decreases in losses across clusters and scenarios for 2025, following the hourly patterns. For 2040, all clusters have higher losses than the base case in the High Electrification scenario, and all clusters have lower losses in the High PV scenario. In the Boundary scenario, losses increase across clusters due to the dominance of EV charging load over PV production.

We extrapolate feeder-level results in Table A.2 for all IOU service territories by applying the same scaling factors described in section C.3 (SI) and by multiplying the average hourly losses by 8,760 to extend these estimates to a whole year (Table 8). These results will then be used in the following section to provide a first-order estimate of the economic impacts.

	Annual Change in Losses Relative to Base (MWh)							
	<b>High Electrification</b>		Hig	h PV	Boundary			
Cluster	2025	2040	2025	2040	2025	2040		
CL1	41	709	-656	-1,264	-710	10,348		
CL2	306	4,478	-1,817	-3,837	-1,651	28,556		
CL3	276	10,289	-2,606	-8,310	-2,861	88,606		
CL4	2,005	19,655	-8,739	-13,278	-856	40,458		

Table 8. Aggregate change in losses relative to base case for all IOUs (MWh)

CL5	985	14,640	-8,310	-17,794	-6,669	39,308
CL6	0	55	-267	-401	-392	129

## 5.2 Generation technical impacts

Hourly net demand for each scenario scaled at the state-level was produced as an input to the capacity expansion and production cost modules of Aurora. Aurora produces optimal capacity expansion decisions, which are tested on an hourly basis employing the production cost module. The resulting electricity prices are input on a demand forecasting model that considers price elasticity, and the resulting demand is input back into Aurora. This modeling system is solved iteratively until equilibrium is reached. The simulation then reflects the incremental generation investment needs and annual costs to meet those demand levels for years 2025 and 2037<sup>3</sup>. Details of the modeling approach and assumptions can be found in [37].

Table 9 reports resulting differences in capacity additions under each scenario over the long-term. Natural gas simple-cycle combustion turbines (SCCT) are deployed in larger amounts in scenarios with higher penetration of electric vehicles, and lower amounts in scenarios with higher PV penetration. The Boundary scenario—dominated by EV adoption—requires more than three times the incremental capacity of SCCTs compared to the Base case despite having only 50% higher peak demand. In contrast, capacity additions of natural gas combined-cycle combustion turbines (CCCT) remain relatively constant across scenarios. The significant adoption of SCCT in the Boundary scenario reflects the flexibility and resource adequacy demands that large swaths of coincident EV charging may impose on the power system.

	Incremental Installed Capacity (MW) Natural Gas:					
Scenario	Natural Gas: Simple Cycle Combustion Turbine	Wind	Solar			
Base	4,971	6,034	5,696	579		
High Electrification	6,214	5,748	7,000	1,278		
High PV	3,879	6,330	2,385	414		
High PV and Storage	3,960	6,338	2,384	316		
Storage	4,987	6,010	5,766	579		
Boundary	16,959	7,360	4,030	55		

Table 9. Utility-scale resource mix by scenario in year 2037

Wind and solar adoption is substantially higher in the High Electrification scenario compared to any other scenario. This is may be due to coincidence between solar and wind production and EV charging patterns. Higher DER PV adoption in scenarios 3 and 4 correlates with lower wind and solar adoption than the Base case. This is explained by DER PV reducing the capacity value of solar PV given the high

<sup>&</sup>lt;sup>3</sup> This is the latest year available in the Aurora modeling implementation available to us, hence it is being used as equivalent to 2040 for our purposes.

production correlation of both resources.

## 6. Economic impacts of DER across the power system

The costs and benefits of DER are determined separately for the three major components of the power system: (1) generation, (2) transmission, and (3) distribution. Generation cost impacts come directly from the Aurora capacity expansion results. Transmission cost impacts are estimated based on peak demand increase. Distribution cost impacts are based on the results described in the previous section.

## 6.1 Distribution

There are three cost components tracked for the integration of DER into the distribution system: (1) voltage regulation, (2) line loading, and (3) line energy losses.

## 6.1.1 Voltage regulation

Results discussed earlier show that voltage issues are a relatively minor issue across scenarios and that, in some cases, they are driven by the high voltage set point at the substation load tap changer. In this study, we assume that smart inverters are a standard feature in PV systems deployed within every scenario presented. Accordingly, we find that voltage issues for all scenarios can be mitigated by a combination of load tap changer (LTC) adjustments and smart inverter use with PV systems. Consequently, simulations using a combination of volt-var control at PV systems and adjustment of substation LTC result in no voltage issues in the short and long term. This approach and result is consistent with similar studies on management of voltage issues due to rooftop solar adoption (see e.g. [38]).

The no-cost result for voltage regulation is based on the assumption that LTC is available and adjusted in the IOU-operated electricity substations across Indiana. Unfortunately, we do not have information confirming the reasonableness of this assumptions. For this reason, we include a cost to retrofit half of the existing substations with LTC—assuming the remainder already have LTC installed.

Researchers report that it costs \$310,000 per substation to implement LTC based on a Northeastern U.S. utility [38]. We adjust this cost down by 25% to \$232,500 based on information from two Indiana utilities. We estimate there are ~1,000 substations serving distribution customers across the Indiana territory. It will cost ~\$235 million, or an annual equivalent of \$20 million, to retrofit all of these substations. In the end, we assume that half of the substations need the LTC retrofit resulting in an annual cost of ~\$10 million.

## 6.1.2 Line loading

Line loading was addressed by manually replacing conductors in underground and overhead line segments as needed. We re-ran simulations for the affected cluster-scenario combinations to verify that the re-conductoring effectively solved line overloading. Tables A.5 to A.7 (SI) include the segment-by-segment details for this re-conductoring process. The lengths of upgraded circuits are reported in Table 10.

Table 10. Length of re-conductored segments by material and cluster

	Underground	Cable Length (feet)	Overhead Li	ne Length (feet)
Cluster	Copper	Aluminum	Copper	Aluminum
3	0	0	0	3,634
4	57	0	2,386	0
5	172	0	0	6,461

We monetize re-conductoring using costs per foot of conductor as reported in two sources. First, two of the three IOUs with representative feeders reported costs of \$95/ft and \$80/ft for overhead and underground line re-conductoring, respectively. The overhead costs include replacing supporting structures to bear additional conductor weight. Second, an NREL cost study reported low, medium, and high costs of \$130/ft, \$173/ft, and \$258/ft, respectively.

We use the preceding cost information to estimate costs based on four re-conductoring "steps" that depend on the ampacity difference between the original and replaced conductor. Each step reflects a 15% increase in conductor ampacity. We assume that the lower cost applies to the first step, and the highest cost to the fourth step. Underground cables are upgraded in a single step, so we use the \$80/ft reported by the utility. Finally, we use a 50% cost adder for copper conductors assuming that all costs are for aluminum conductors. Feeder level results are escalated to the aggregate IOU level using the scaling factors described earlier in this manuscript (Table 11).

	Feeder Costs (milli	on \$2017)	Aggregate Costs for All IOUs (million \$20)		
Cluster	High Electrification	Boundary	High Electrification	Boundary	
3		\$0.396		\$297.0	
4	\$0.271	\$0.412	\$147.4	\$223.8	
5		\$0.973		\$306.5	

Table 11. Feeder-level and aggregate costs for line loading by scenario and cluster

Re-conductoring was only required for clusters 3, 4 and 5 and for the High Electrification and the Boundary scenarios. We estimate about \$150 million in upgrade costs for feeders in cluster 4 in the High Electrification scenario, and roughly \$820 million in investments for the Boundary scenario. These investment values correspond to approximate annualized costs of \$12.5 million for the High Electrification scenario and \$70 million for the Boundary scenario.

It is important to acknowledge that this linear segment-by-segment upgrade method is just one of the ways in which utilities address real line loading issues in their systems. Our assessment uses individual segment upgrades, usually called an "incremental line upgrade", largely due to data availability and resource constraints. Another example of an incremental upgrade not employed in this study is adding phases to a single-phase circuit to increase its capacity. Furthermore, in some situations, poles will need to be replaced along supporting structures and conductors in a "major line upgrade". In some cases these methods will be insufficient and utilities may be required to build additional feeder sections and reconfigure feeders to offload affected circuits. Regulators, utilities, and/or other stakeholders should consider sponsoring a more detailed line-loading study under different DER adoption pathways.

#### 6.1.3 Energy losses

Distribution system energy losses is energy that a utility procured, but could not deliver to end-use customers. Estimating the cost of these losses entails using an average wholesale market delivery cost to value the energy losses first reported in Table 8. We use the generation and transmission costs— reported in dollars per MWh— from the generation production cost model to monetize energy losses under each scenario relative to the base case (see Table 12). The cost of energy-related losses in the High PV and High PV and Storage scenarios are identical and there is no difference between the Base case and Storage scenario.

	202	25	2040			
	Wholesale Electricity Cost	Cost of Energy	Wholesale Electricity Cost	Cost of Energy		
Scenario	(¢/kWh)	\$2017)	(¢/kWh)	\$2017)		
High Electrification	4.23¢	\$0.15	5.40¢	\$2.69		
High PV	4.05¢	-\$0.91	5.21¢	-\$2.34		
High PV and Storage	4.05¢	-\$0.91	5.21¢	-\$2.34		
Storage	4.16¢	\$0	5.32¢	\$0		
Boundary	4.31¢	-\$0.57	6.05¢	\$12.55		

Table 12. Changes in the cost of energy losses relative to the base case

By 2025, the economic impact of energy losses under increased DER adoption in energy losses is modest, ranging from an additional cost of \$150,000/year in the High Electrification scenario to savings of almost \$1 million per year in the two High PV scenarios. The economic impact becomes more over the long-term. The High PV scenarios save over \$2 million in distribution-related energy losses compared to the Base case, while the Boundary scenario has an additional \$12.5 million in energy loss-related costs compared to the Base case.

#### 6.2 Transmission

Transmission expansion is not modeled directly in the comprehensive study. We estimate the impact of DER on transmission costs by calculating an incremental transmission expansion cost per MW transmitted during peak hours in 2025 and 2037 and multiplying this value by the peak demand in each adoption scenario.

We estimated the incremental transmission expansion costs by comparing the revenue requirements for the reference scenario in the ratemaking model both with and without incremental transmission expenditures. These expenditures include the return on investment and depreciation of all future capital expenditures, but not from the current rate base, and future transmission system O&M costs. These costs were translated to a dollar per peak MW basis for the revenue requirements in 2025 and 2037 for the reference scenario. This process produces an incremental transmission cost of \$55,821 per peak MW in 2025 and \$68,896 per peak MW in 2037 that are reasonable approximations for expansion costs in the transmission system.

We apply these values to the statewide peak demand by scenario to estimate transmission costs and

calculate the difference from the Base case (see Table 13 for cents per kWh costs and Table A.4 in SI for total costs). DER impact is relatively modest in all but the Boundary scenario, with savings of 3 cents per MWh in the High PV and High PV and Storage scenarios in 2025 to an increase of 57 cents per MWh in the High Electrification scenario in 2037. These figures translate to differences in the -0.3% to 4.7% range.

	Cost Change w Base Case	ith Respect to (¢/kWh)	Annual Cost Change with Respect to Base Case (million S		
	2025	2037	2025	2037	
High Electrification	0.01¢	0.06¢	\$15.8	\$91.3	
High PV	0.00¢	0.01¢	-\$32.4	-\$71.9	
High PV and Storage	0.00¢	0.01¢	-\$32.4	-\$70.6	
Storage	0.00¢	0.00¢	\$0	\$0.01	
Boundary	0.07¢	0.64¢	\$27.5	\$734	

Table 13. Changes to incremental transmission costs relative to the base case

The Boundary scenario has the highest cost difference for both planning horizons. Transmission costs are almost 7% higher in 2025 and up to 53% higher in 2037. This is explained due to the peak demand levels of this scenario, which at 31.8 GW in 2037 are roughly 50% higher than the 21.1 GW in the Base scenario.

## 6.3 Generation

We report four components of generation costs produced by the simulations: (1) annualized capital costs, (2) fixed costs, (3) fuel costs, and (4) non-fuel variable costs (usually O&M).

In the short-term, all scenarios, including the Boundary case, exhibit similar costs relative to the Base case (Fig. 6). However, over the long-term, the cost differences associated with increased adoption levels become more evident. Scenarios with relatively higher adoption of PV (High PV and High PV and Storage) have 8% lower costs relative to the Base case, largely driven by reduced capital and fixed costs. Costs are roughly 3% higher relative to the Base case in the High Electrification scenario, likely driven by EV charging taking place in the middle of the day. This is supported by the much higher adoption of utility-scale solar PV—whose production peaks midday—in this scenario compared to any other scenario (e.g. twice as much as the Base case).

The Battery Storage scenario is basically identical to the Base case. This is, in part, due to the relatively small levels of adoption of DER storage. However, it also suggests that when customers manage their DER storage without following wholesale market signals, their decisions do not necessarily benefit the system through lower peak demand needs. Finally, the Boundary scenario has ~12% higher costs than the Base case, driven by the strong demand growth of EV charging. It is important to note that the Boundary scenario is serving a 50% higher peak demand than the Base case.

These results suggest that DER adoption, especially PV, could create significant costs savings in both energy and capacity for the Indiana power system. The High PV scenario has 3% higher fuel costs, but 30% lower annual capacity costs compared to the Base case by 2040. In contrast, the higher demand







## 6.4 Economic and rate impacts of DER adoption

Table 14 shows the incremental combined economic impact of increased DER adoption relative to the Base case in cents per kWh (see Table A.3 in SI for absolute cost impacts). Costs are reported by scenario and for the three segments of the power system: generation, transmission, and distribution.

Table 14. Overall incremental economic impact of DER adoption by scenario and power system segment relative to the base case (2017 cents/kWh)

	2025 A	2025 Annual Cost Change Relative to Base			2040 Annual Cost Change Relative to Ba			
Scenario	Gen.	Trans.	Dist.	Total	Gen.	Trans.	Dist.	Total
High Electrification	0.11¢	0.02¢	0.01¢	0.14¢	0.25¢	0.11¢	0.03¢	0.39¢
High PV	-0.34¢	-0.04¢	0.01¢	-0.37¢	-0.64¢	-0.09¢	0.01¢	-0.72¢
High PV and Storage	-0.34¢	-0.04¢	0.01¢	-0.37¢	-0.63¢	-0.09¢	0.01¢	-0.72¢
Storage	0.00¢	0.00¢	0.01¢	0.02¢	0.00¢	0.00¢	0.01¢	0.02¢
Boundary	-0.03¢	0.04¢	0.01¢	0.03¢	0.96¢	0.93¢	0.12¢	2.01¢

There are relatively modest economic impacts of DER adoption for all scenarios in the short term. Over the long term, impacts range from 0.7 ¢/kWh in savings for the High PV scenarios to 2 ¢/kWh in additional costs for the Boundary scenario, all relative to the Base case. The largest cost impacts are observed in the generation sector, with nearly 90% of the cost savings occurring in this segment for the High PV scenarios. Distribution-related cost impacts from DER adoption are generally the smallest among the power system segments studied, ranging 1% to 10% of the overall cost change under any given scenario.

Rate impacts of these incremental costs are reported in Table 15. This assessment employs the ratemaking model using the existing rate base and the incremental cost changes reported in Table A.3 in SI. The ratemaking model takes into consideration DER-driven investments reported in this study and non DER-driven investments that are part of utilities' investment plans, in addition to increase or reduction in retail sales and peak demand depending on the scenario. Details of the ratemaking model can be found in [37,39].

	2025 Rate Change Relative to Base				2040 Rat	te Change Re	elative to E	Base
Scenario	Residential	Commercial	Industrial	Average	Residential	Commercial	Industrial	Average
High Electrification	0.25¢	0.24¢	0.19¢	0.22¢	-0.03¢	0.05¢	0.14¢	0.06¢
High PV	-0.06¢	-0.10¢	-0.19¢	-0.13¢	1.01¢	0.73¢	0.23¢	0.59¢
High PV and Storage	-0.06¢	-0.10¢	-0.19¢	-0.13¢	1.00¢	0.71¢	0.22¢	0.58¢
Storage	0.00¢	0.00¢	0.00¢	0.00¢	0.05¢	0.05¢	0.01¢	0.03¢
Boundary	0.52¢	0.47¢	0.18¢	0.35¢	1.88¢	1.96¢	1.46¢	1.70¢

Table 15. Impact of DER adoption on electricity rates by scenario and customer type (2017 cents/kWh)

In contrast to the incremental costs reported earlier, average all-in rates increase for all scenarios in the long term. In the High PV scenarios this is due to the reduction in sales that needs to be compensated with higher rates for utilities to recover their fixed costs. In the other scenarios this is compounded with the need for incremental generation and transmission infrastructure to meet increased peak demand. Overall, the average changes in rates are relatively modest in the non-Boundary scenarios. Rates are expected to decrease in the short-term in the High PV scenarios (~1.6%) and to increase in the High Electrification and Boundary scenarios (~2%). In the long-term, rates increase between 0.2% to 2.5% in non-Boundary scenarios and 14.5% in the Boundary scenario due to very high unmanaged EV load. This result suggests distributional issues of EV owners potentially imposing rate increases on non-EV owners.

## 7. Conclusion

This study explores the joint impacts across the power system of DER technologies. We identify six adoption scenarios that combine deployment levels of rooftop solar (PV), electric vehicle charging (EV), and battery storage in residential and commercial customers connected to representative feeders in Indiana by 2025 and 2040. Indiana is a good proxy for many U.S. states with low current DER adoption but potentially high future growth.

#### This paper uses

Methodologically, this is the first paper that develops and applies a sequential integrative framework that rigorously identify technical, cost, and rate impacts across all segments in the power system. The sequential approach operates in two stages. First, it simulates DER impacts in the distribution system using a clustering technique to identify six representative feeders. Second, it suggests an approach to scale and aggregate resulting net demand to produce the inputs for a transmission-generation analysis. Each stage includes estimation of technical and cost impacts using industry-standard power flow and capacity expansion tools. Finally, the technical and economic impacts of all three segments – distribution, transmission, and generation – are input into a rate making model that jointly processes the changes in retail energy sales and demand with the changes in investment and operational costs across segments.

The scenario-based sequential methodology proposed in this paper integrates methods from several strands of literature that have individually analyzed DER impacts on different segments of the power system. The method proposed is particularly well suited for utilities, system planners, regulators, and other stakeholders that require a blueprint to conduct state- and national-level analyzes of the penetration of DER and its impacts on ratepayers. The method develops simple but insightful scenarios for DER adoption and operation that are easily reconstructed with own data and that represent a wide range of future performance possibilities for these technologies. The data and tools required to implement the method proposed in this paper are readily available to the stakeholders mentioned before, maximizing its applicability.

This paper develops an application of the method based on data provided by the five IOUs operating in the State of Indiana. These results show the applicability of the method and exemplify its implementation. In addition, they suggest interest system-wide impacts of DER and their interactions that would apply to most U.S. states and other regions in the world. We find that primary voltage distribution systems are well situated to absorb moderate to large DER load and production. Optimal range voltage violations appear in at least one feeder node on 159 of the 3,456 simulated hours and acceptable range violations in 17 simulated hours. Most voltage violations occur in the longest feeder (e.g. clusters 3 and 4) and seem to be driven primarily by DER solar and EV operation. Line loading is the most common technical impact on distribution systems: between 1% and 8% of circuit segments are overloaded across cluster feeders 3, 4 and 5, especially in the "stress test" or Boundary scenario. Distribution system impacts could be mitigated with a mix of volt-var regulation from smart inverters, reconductoring, and load tap changer implementation and control in substations. It is estimated that the economic impact of these mitigation strategies in the primary voltage system by year 2040 would be between 0.01¢/kWh and 0.03¢/kWh in the five standard scenarios, and 0.12¢/kWh for the Boundary scenario. However, the secondary voltage system - transformers and circuits - may require more complex circuit segmentation and expensive reconductoring on underground circuits that are not captured in this study. This is a further area of research.

Out of the three power system components, generation is the most impacted due to needed peak demand capacity that stem from generally tighter planning margins compared to distribution and transmission. Unmanaged EV charging drive gas combustion turbine adoption for peak demand, at a

rate of roughly 1 MW of gas turbine capacity per MW of EV demand. Higher DER PV adoption displaces utility-scale wind and solar PV capacity by 20%-50% and 30%-90% respectively, compared to the BAU scenario. Higher peak capacity could be mitigated through imports for utilities that are members of RTO/ISO, which would in turn strain the transmission system. These result demonstrate the relevance of integrative results that identify the components of the power system that are more impacted. They also show the substantial impacts of DER adoption in the BPS and suggest that managed EV charging should be prioritized. An analysis that includes regional modeling and managed EV charging was outside the scope of this work but it is a worthwhile research avenue.

Plausible scenarios for EV, PV, and battery storage adoption by residential and commercial customers in Indiana suggest that PV and battery storage will have low to moderate impact. EV adoption will likely be the DER that will produce the highest technical and economic impacts, especially with unmanaged charging or in the absence of time-sensitive rates. Results show modest long-term increase in rates up to 2.5% in non-Boundary scenarios, but up to 14% increase in the Boundary scenario due to EV impacts in the BPS. DER PV could reduce rates in the short-term, but reduced retail sales lead to higher rates in the long-term. Regulators will need to assess whether to react to the distributional consequences of allin rates increasing for non-EV owners.

## 8. Acknowledgments

We would like to acknowledge helpful feedback provided by the staff at the Indiana Utility Regulatory Commission: Brad Borum, Bob Morgan, and Dale Thomas. We would also like to acknowledge staff at the five utilities that provided critical data, useful information, and constructive feedback: Alison Becker, Mark Pallok, and Chad Richwalski (NIPSCO); Matt Fields, Jake Allen, and Justin Sufan (IP&L); Justin Hage, Matthew Rice, Heather Watts, Jason Stephenson, and Ryan Abshier (Vectren); Regiana Sistevaris, Marc Lewis, and Rod Walter (I&M); and Beth Herriman and Nancy Gay (Duke Energy). Finally, we acknowledge the useful comments and suggestions of the three journal reviewers and editor.

Funding: The work described in this study was funded by the Indiana Utility Regulatory Commission under Strategic Partnership Projects Agreement No. FP00009789.

## 9. References

- [1] Carvallo JP, Zhang N, Murphy SP, Leibowicz BD, Larsen PH. The economic value of a centralized approach to distributed resource investment and operation. Appl Energy 2020;269:115071. https://doi.org/10.1016/j.apenergy.2020.115071.
- [2] Cale J, Palmintier B, Narang D, Carroll K. Clustering distribution feeders in the Arizona Public Service territory. 2014 IEEE 40th Photovolt. Spec. Conf. PVSC, 2014, p. 2076–81. https://doi.org/10.1109/PVSC.2014.6925335.
- [3] Broderick RJ, Williams JR. Clustering methodology for classifying distribution feeders. 2013 IEEE 39th Photovolt. Spec. Conf. PVSC, 2013, p. 1706–10. https://doi.org/10.1109/PVSC.2013.6744473.
- [4] Schneider K, Chen Y, Engle D, Chassin D. A Taxonomy of North American Radial Distribution Feeders. 2009.

- [5] Brown RE, Freeman LAA. Analyzing the reliability impact of distributed generation. 2001 Power Eng. Soc. Summer Meet. Conf. Proc. Cat No01CH37262, vol. 2, 2001, p. 1013–8 vol.2. https://doi.org/10.1109/PESS.2001.970197.
- [6] PNNL. Summary of Electric Distribution System Analyses with a Focus on DERs. 2017.
- [7] Seguin R, Woyak J, Costyk D, Hambrick J, Mather B. High-Penetration PV Integration Handbook for Distribution Engineers. Golden, Colorado: NREL; 2016.
- [8] EPRI. Planning Methodology to Determine Practical Circuit Limits for Distributed Generation. Palo Alto, California: EPRI; 2010.
- [9] Schwarz M, Auzépy Q, Knoeri C. Can electricity pricing leverage electric vehicles and battery storage to integrate high shares of solar photovoltaics? Appl Energy 2020;277:115548. https://doi.org/10.1016/j.apenergy.2020.115548.
- [10] Rugthaicharoencheep N, Auchariyamet S. Technical and Economic Impacts of Distributed Generation on Distribution System. Int J Electr Comput Eng 2012;6:385–9.
- [11] NREL. Methods for Analyzing the Benefits and Costs of Distributed Photovoltaic Generation to the U.S. Electric Utility System. 2014.
- [12] RMI. A Review of Solar PV Benefit and Cost Studies. 2013.
- [13] Cohen MA, Kauzmann PA, Callaway DS. Economic Effects of Distributed PV Generation on California's Distribution System. Energy Institute at Haas; 2015.
- [14] Orrell A, Homer J, Preziuso D, Somani A. Illinois Distributed Generation Rebate Preliminary Stakeholder Input and Calculation Considerations. Pacific Northwest National Laboratory; 2018.
- [15] Tor OB, Guven AN, Shahidehpour M. Promoting the Investment on IPPs for Optimal Grid Planning. IEEE Trans Power Syst 2010;25:1743–50. https://doi.org/10.1109/TPWRS.2009.2039947.
- [16] Tor OB, Guven AN, Shahidehpour M. Congestion-Driven Transmission Planning Considering the Impact of Generator Expansion. IEEE Trans Power Syst 2008;23:781–9. https://doi.org/10.1109/TPWRS.2008.919248.
- [17] Barbose G. Putting the Potential Rate Impacts of Distributed Solar into Context. 2018. https://doi.org/10.2172/1469160.
- [18] Cappers P, Satchwell A, Gorman W, Reneses J. Financial Impacts of Net-Metered Distributed PV on a Prototypical Western Utility's Shareholders and Ratepayers 2019:19.
- [19] ERCOT. DISTRIBUTED ENERGY RESOURCES (DERS) Reliability Impacts and Recommended Changes. ERCOT; 2017.
- [20] NERC. Distributed Energy Resources: Connection Modeling and Reliability Considerations. 2017.
- [21] NATF. NATF Reference Document: Distributed Energy Resource Modeling and Study Practices. North American Transmission Forum; 2018.
- [22] Zhang X, Strbac G, Teng F, Djapic P. Economic assessment of alternative heat decarbonisation strategies through coordinated operation with electricity system – UK case study. Appl Energy 2018;222:79–91. https://doi.org/10.1016/j.apenergy.2018.03.140.
- [23] Bloess A. Impacts of heat sector transformation on Germany's power system through increased use of power-to-heat. Appl Energy 2019;239:560–80. https://doi.org/10.1016/j.apenergy.2019.01.101.
- [24] Aunedi M, Pantaleo AM, Kuriyan K, Strbac G, Shah N. Modelling of national and local interactions between heat and electricity networks in low-carbon energy systems. Appl Energy 2020;276:115522. https://doi.org/10.1016/j.apenergy.2020.115522.
- [25] Sarstedt M, Garske S, Blaufuß C, Hofmann L. Modelling of Integrated Transmission and Distribution Grids based on Synthetic Distribution Grid Models. 2019 IEEE Milan PowerTech, 2019, p. 1–6. https://doi.org/10.1109/PTC.2019.8810823.

- [26] Garske S, Blaufuß C, Sarstedt M, Hofmann L. Reactive power management analyses based on generic distribution grid models. 2017 IEEE PES Innov. Smart Grid Technol. Conf. Eur. ISGT-Eur., 2017, p. 1–6. https://doi.org/10.1109/ISGTEurope.2017.8260248.
- [27] Pudjianto D, Aunedi M, Djapic P, Strbac G. Whole-Systems Assessment of the Value of Energy Storage in Low-Carbon Electricity Systems. IEEE Trans Smart Grid 2014;5:1098–109. https://doi.org/10.1109/TSG.2013.2282039.
- [28] Carvallo J, Taneja J, Callaway D, Kammen DM. Distributed Resources Shift Paradigms on Power System Design, Planning, and Operation: An Application of the GAP Model. Proc IEEE 2019:1–17. https://doi.org/10.1109/JPROC.2019.2925759.
- [29] Clack C, Choukulkar A, Cote B, McKee S. Why Local Solar For All Costs Less: A New Roadmap for the Lowest Cost Grid. Vibrant Clean Energy, LLC; 2020.
- [30] Arentsen MJ, Künneke RW. Economic organization and liberalization of the electricity industry: In search of conceptualization. Energy Policy 1996;24:541–52. https://doi.org/10.1016/0301-4215(96)00044-4.
- [31] CYME. CYME Distribution System Analysis Brochure 2018.
- [32] EPRI. The Integrated Grid: Realizing the Full Value of Central and Distributed Energy Resources. 2014.
- [33] Frick N, Schwartz L, Taylor-Anyikire A. A Framework for Integrated Analysis of Distributed Energy Resources: Guide for States 2018.
- [34] Shenot J, Linvill C, Dupuy M, Brutkoski D. Capturing More Value from Combinations of PV and Other Distributed Energy Resources. Montpelier, VT: Regulatory Assistance Project (RAP); 2019.
- [35] Nassuato S, Magistrati G, Marchegiani G, Brivio C, Delfanti M, Falabretti D, et al. Distributed Storage for the Provision of Ancillary Services to the Main Grid: Project PRESTO. Energy Procedia 2016;99:182–93. https://doi.org/10.1016/j.egypro.2016.10.109.
- [36] Denholm P, Margolis R, Palmintier B, Barrows C, Ibanez E, Bird L, et al. Methods for Analyzing the Benefits and Costs of Distributed Photovoltaic Generation to the U.S. Electric Utility System. 2014. https://doi.org/10.2172/1159357.
- [37] Phillips TA, Velástegui MA, Gotham DJ, Lu L, Wu F, Nderitu DG, et al. Indiana Electricity Projections: The 2019 Forecast. Purdue University; 2019.
- [38] Horowitz KAW, Ding F, Mather BA, Palmintier BS. The Cost of Distribution System Upgrades to Accommodate Increasing Penetrations of Distributed Photovoltaic Systems on Real Feeders in the United States. 2018. https://doi.org/10.2172/1432760.
- [39] Hadley SW. ORFIN: An electric utility financial and production simulator. Oak Ridge National Lab., TN (United States); 1996. https://doi.org/10.2172/226026.